Author's response to reviews

Title: Asystole Following Positive Pressure Insufflation of Right Pleural Cavity

Authors:

Kari M Forde-Thielen (ford0064@umn.edu)
Mojca R Konia (konia012@umn.edu)

Version: 3 Date: 17 February 2011

Author's response to reviews:

From:
Mojca Remskar Konia, M.D.
University of Minnesota
Department of Anesthesiology
Box 294, B515 Mayo Memorial Bldg.
420 Delaware Street, SE
Minneapolis, MN 55455
Phone: 612 624 9990
Fax: 612 626 2363
E-mail: konia012@umn.edu

Date: 2/17/2011

Thank you very much for considering our case report for publication in your journal.

Please find attached a copy of a revised case report entitled “Asystole Following Positive Pressure Insufflation of Right Pleural Cavity ”.

Based on the comments we have received from the Science Editorial Office, we have restructured the abstract and included ethnicity of the patient as suggested.

Sincerely,
Mojca Remskar Konia, MD